\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{www.ijrps.com}{\color{blue}\underline{\smash{www.ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage{float}

\begin{document}


\def\RunningAuthor{Kannappan Poornima et al.}
\firstPage{4111}
\articleType{Original Article}
\receivedDate{01 Mar 2020}
\acceptedDate{09 May 2020}
\revisedDate{25 Apr 2020}
\journalVolume{2020, 11}
\journalIssue{3}
\journalDoi{ijrps.v11i3.2614}
\copyrightYear{2020}
\def\authorCount{3}
\def\affCount{1}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{Acute and Sub chronic Toxicological evaluation of ethanolic leaf extract of \textit{Sida acuta} Burm.F in Wistar albino rats by analyzing biochemical, heamatological and histochemical parameters}
\author{Perumalsamy~Muneeswari,  
            Sri~Rashmy~Madathil, 
            Kannappan~Poornima\textsuperscript{*}~\\[5pt]{Department of Biochemistry, Karpagam Academy of Higher Education, Coimbatore - 641 021, Tamil Nadu, India}}

\begin{abstract}
\textit{Sida acuta }burm.f belongs to \textit{Malvaceae, }the mallow family and enjoys tropical and pantropical distribution. The plant is usually known as wireweed in the countryside, and it is highly medicinally valued traditionally and ethnobotanically promised. The present study is concerned with acute and sub-chronic toxicity evaluation of ethanolic extract of \textit{Sida acuta} Burm.f leaves in Wistar albino rats. Acute toxicity evaluation was conducted for 14 days. Acute doses of 100, 250, 500, 1000 and 2000 mg/kg BW were administrated to test groups of animals under consideration on the first day of experimental evaluation with three animals in each of total six groups along with control. For the remaining 13 days, animals were observed for noted behavioural changes and body weight were recorded respectively for 7\ensuremath{^{th}} and 14\ensuremath{^{th}} day of experimental analysis. At the end of the trial period, all the animals were euthanised, and various biochemical parameters and histopathological examination were carried out to assess the toxicity of extract. The present study revealed that the ethanolic extract of \textit{Sida acuta } Burm.f leaves is non-toxic up to 2000mg/kg body weight. Subchronic toxicity evaluation was conducted for 28 days with several doses 100, 200, 300, 400 and 500mg/kg BW. Control rats without any treatment were maintained during the entire period of experimental analysis. The results of subchronic toxicity parameters indicate no significant changes to the biochemical parameters (glucose, urea, uric acid, creatinine, AST, ALT and Cholesterol) haematological and histopathological observation in comparison to the control groups. Based on subchronic toxicity parameters data, effective doses (200 and 400mg/kg BW) is determined for further cancer (colon) study in Wistar albino rats.
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Sida acuta Burm.f,\newline Toxicity,\newline Lethal dose
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Kannappan~Poornima~\\ Phone:\ +91-0422-6453777~\\  Email:\ poorniovya@gmail.com
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{www.ijrps.com}{www.ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
Herbal medicines and medicinal herbs are receiving considerable importance worldwide. According to the report of the WHO (World Health Organization), about 80\% of people in developing countries depend on herbal medicine for their preliminary health care due to its easy availability, low cost and convincing results. In developed countries also, the public interest in herbal prescription has dramatically increased \unskip~\citep{798243:19046235}. The bioactive principles of medicinal plants constitute the essential ingredients for drug development and therapeutics. Researchers have shown that medicinal plants have antimicrobial, analgesic, anti-candidiasis activities, hypotensive activity, antidiabetic, antiproliferative, anti-inflammatory and antioxidant \unskip~\citep{798243:19045978,798243:19046237,798243:19046039}. 

However, although medicinal plants have several therapeutic virtues, they have to be adequately screened for toxicological parameters in a biological system as some of them are at risk of toxicity. Several researchers have pointed out the potential toxicity, as well as the risks associated with the use of certain species of plants and vegetables \unskip~\citep{798243:19046042,798243:19046043}. Seen, the dangers of toxicity, studies on the safety and effectiveness of medicinal plants have become one of the main concerns to guarantee the use \unskip~\citep{798243:19046232,798243:19046236}.

\textit{Sida acuta}, is an erect perennial shrub found throughout the warmer provinces of India and Nepal. In traditional and indigenous system of medicine, it is used to treat gonorrhoea, elephantiasis and ulcers and is claimed to have aphrodisiac properties. The juice of the root is applied to wounds. The whole plant is used to treat snakebite, and it lessened the hemorrhagic effect of \textit{Bothrops atrox} venom \unskip~\citep{798243:19046046,798243:19046231} \textit{. } Alkaloid cryptolepine isolated from \textit{Sida acuta} was proved to be responsible for the antiplasmodial activity of the present species \unskip~\citep{798243:19046041,798243:19047076}. The aerial part of the plant is the most frequently used part. In Central America, the plant is used to treat renal inflammation, asthma, colds, fever, ulcers, worms and headache, \unskip~\citep{798243:19047077}. The present study aims to investigate the toxicological evaluation of the ethanolic extract of \textit{Sida acuta} burm.f leaves using Wistar albino rats. 


\begin{table*}[!htbp]
\caption{\boldmath {Hematological parameters of sub chronic toxicity study } }
\label{tw-030d44184b0b}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1578\linewidth-2\tabcolsep}p{\dimexpr.1222\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Parameters   & Group I~& Group II & Group III & Group IV & Group V & Group VI\\
\tblmidrule 
Total WBC Count Cells/Cu.mm  &
  8,201 \ensuremath{\pm} 0.4\ensuremath{^{a}} &
  8,251 \ensuremath{\pm} 0.5\ensuremath{^{a }} &
  8,204 \ensuremath{\pm} 0.52\ensuremath{^{a}} &
  8,200 \ensuremath{\pm} 0.57\ensuremath{^{a}} &
  8,201 \ensuremath{\pm} 0.13\ensuremath{^{a }} &
  8,201 \ensuremath{\pm} 0.3\ensuremath{^{a }}\\
Total RBC Count Million/Cu.mm  &
  4 \ensuremath{\pm} 0.25\ensuremath{^{a}} &
  3.9 \ensuremath{\pm} 0.15\ensuremath{^{a}} &
  3.9 \ensuremath{\pm} 0.16\ensuremath{^{a }} &
  4 \ensuremath{\pm} 0.72 \ensuremath{^{a}} &
  4 \ensuremath{\pm} 0.08\ensuremath{^{a}} &
  4 \ensuremath{\pm} 0.07\ensuremath{^{a }}\\
Hemoglobin g/dL  &
  13 \ensuremath{\pm} 0.40\ensuremath{^{a }} &
  13 \ensuremath{\pm} 0.1\ensuremath{^{a}} &
  13 \ensuremath{\pm} 0.40\ensuremath{^{a}} &
  13 \ensuremath{\pm} 0.35\ensuremath{^{a}} &
  13\ensuremath{\pm} 0.10\ensuremath{^{a}} &
  14\ensuremath{\pm} 0.36\ensuremath{^{a}}\\
HCT/PCV \%  &
  38 \ensuremath{\pm} 0.60\ensuremath{^{a }} &
  39 \ensuremath{\pm} 0.36\ensuremath{^{a }} &
  40 \ensuremath{\pm} 0.85\ensuremath{^{a }} &
  40 \ensuremath{\pm} 0.76\ensuremath{^{a }} &
  38 \ensuremath{\pm} 0.2\ensuremath{^{a }} &
  39 \ensuremath{\pm} 0.26\ensuremath{^{a }}\\
Platelet Count Lakhs/Cu.mm  &
  4 \ensuremath{\pm} 0.22\ensuremath{^{a }} &
  3.9\ensuremath{\pm} 0.25\ensuremath{^{a}} &
  3.8\ensuremath{\pm} 0.29\ensuremath{^{a }} &
  3.8 \ensuremath{\pm} 0.35\ensuremath{^{a }} &
  4 \ensuremath{\pm} 0.31\ensuremath{^{a }} &
  3.9 \ensuremath{\pm} 0.16\ensuremath{^{a}}\\
MCV Cu.um  &
  83 \ensuremath{\pm} 1\ensuremath{^{a}} &
  82 \ensuremath{\pm} 1\ensuremath{^{a}} &
  82 \ensuremath{\pm} 1\ensuremath{^{a }} &
  81 \ensuremath{\pm} 1\ensuremath{^{a}} &
  83 \ensuremath{\pm} 1\ensuremath{^{a }} &
  82 \ensuremath{\pm} 1\ensuremath{^{a}}\\
MCH pg  &
  25 \ensuremath{\pm} 1\ensuremath{^{a}} &
  24 \ensuremath{\pm} 0.20\ensuremath{^{a }} &
  25\ensuremath{\pm} 0.25\ensuremath{^{a}} &
  24 \ensuremath{\pm} 0.55\ensuremath{^{a }} &
  25 \ensuremath{\pm} 0.70\ensuremath{^{a }} &
  24 \ensuremath{\pm} 0.1\ensuremath{^{a }}\\
MCHC g/dL  &
  47 \ensuremath{\pm} 1\ensuremath{^{a }} &
  46 \ensuremath{\pm} 0.5\ensuremath{^{a }} &
  48 \ensuremath{\pm} 0.64\ensuremath{^{a }} &
  46\ensuremath{\pm} 0.6\ensuremath{^{a }} &
  45 \ensuremath{\pm} 0.68\ensuremath{^{a }} &
  45 \ensuremath{\pm} 0.36\ensuremath{^{a }}\\
RDW \%  &
  15 \ensuremath{\pm} 0.90\ensuremath{^{a}} &
  15 \ensuremath{\pm} 0.60\ensuremath{^{a }} &
  16 \ensuremath{\pm} 0.40\ensuremath{^{a }} &
  15 \ensuremath{\pm} 0.68\ensuremath{^{a}} &
  15 \ensuremath{\pm} 0.85\ensuremath{^{a }} &
  15\ensuremath{\pm} 0.26\ensuremath{^{a }}\\
MPV Cu.um  &
  7 \ensuremath{\pm} 0.60\ensuremath{^{a }} &
  7 \ensuremath{\pm} 0.45\ensuremath{^{a }} &
  7 \ensuremath{\pm} 0.25  &
  7 \ensuremath{\pm} 0.3\ensuremath{^{a}} &
  8 \ensuremath{\pm} 1\ensuremath{^{a }} &
  7 \ensuremath{\pm} 0.1\ensuremath{^{a }}\\
PCT \%  &
  0.326 \ensuremath{\pm} 0.05\ensuremath{^{a}} &
  0.4 \ensuremath{\pm} 0.17  &
  0.3 \ensuremath{\pm} 0.15  &
  0.3 \ensuremath{\pm} 0.06  &
  0.4 \ensuremath{\pm} 0.14  &
  0.3 \ensuremath{\pm} 0.06 \\
PDW \%  &
  8 \ensuremath{\pm} 0.152\ensuremath{^{a }} &
  9 \ensuremath{\pm} 0.60\ensuremath{^{a}} &
  9 \ensuremath{\pm} 0.68\ensuremath{^{a}} &
  8\ensuremath{\pm} 0.1\ensuremath{^{a }} &
  9 \ensuremath{\pm} 0.20\ensuremath{^{a }} &
  8\ensuremath{\pm} 0.1\ensuremath{^{a }}\\
Polymorphs \%  &
  40 \ensuremath{\pm} 1\ensuremath{^{a }} &
  40 \ensuremath{\pm} 1\ensuremath{^{a}} &
  40 \ensuremath{\pm} 0.2\ensuremath{^{a}} &
  41 \ensuremath{\pm} 0.76\ensuremath{^{a }} &
  41 \ensuremath{\pm} 0.68\ensuremath{^{a }} &
  40 \ensuremath{\pm} 0.76\ensuremath{^{a }}\\
Lymphocytes \%  &
  88 \ensuremath{\pm} 1\ensuremath{^{a}} &
  87 \ensuremath{\pm} 1\ensuremath{^{a }} &
  87 \ensuremath{\pm} 0.3\ensuremath{^{a}} &
  86 \ensuremath{\pm} 0.35\ensuremath{^{a }} &
  89 \ensuremath{\pm} 0.5\ensuremath{^{a}} &
  87\ensuremath{\pm} 1\ensuremath{^{a}}\\
Monocytes \%  &
  5 \ensuremath{\pm} 0.5\ensuremath{^{a }} &
  5 \ensuremath{\pm} 0.76\ensuremath{^{a}} &
  5 \ensuremath{\pm} 0.5\ensuremath{^{a }} &
  5 \ensuremath{\pm} 0.25\ensuremath{^{a }} &
  6 \ensuremath{\pm} 0.76\ensuremath{^{a}} &
  5 \ensuremath{\pm} 1\ensuremath{^{a }}\\
Lymphocyte Count Cells/Cu.mm  &
  3,601 \ensuremath{\pm} 1\ensuremath{^{a}} &
  3,603 \ensuremath{\pm} 3.51\ensuremath{^{a}} &
  3,601 \ensuremath{\pm} 0.7\ensuremath{^{a}} &
  3,600 \ensuremath{\pm} 0\ensuremath{^{a }} &
  3,601 \ensuremath{\pm} 1\ensuremath{^{a}} &
  3,601 \ensuremath{\pm} 1\ensuremath{^{a }}\\
Monocyte Count Cells/Cu.mm  &
  1201 \ensuremath{\pm} 0.76\ensuremath{^{a }} &
  1,202 \ensuremath{\pm} 1.52\ensuremath{^{a }} &
  1301 \ensuremath{\pm} 0.2\ensuremath{^{a }} &
  1,201 \ensuremath{\pm} 1\ensuremath{^{a }} &
  1,201 \ensuremath{\pm} 1\ensuremath{^{a }} &
  1201 \ensuremath{\pm} 0.76\ensuremath{^{a }}\\
Granulocyte Count Cells/Cu.mm  &
  1,401 \ensuremath{\pm} 0.01\ensuremath{^{a}} &
  1,401\ensuremath{\pm} 0.02\ensuremath{^{a}} &
  1,403 \ensuremath{\pm} 0.01\ensuremath{^{a}} &
  1,40 1\ensuremath{\pm} 0\ensuremath{^{a }} &
  1,400 \ensuremath{\pm} 0\ensuremath{^{a}} &
  1,401 \ensuremath{\pm} 0.01\ensuremath{^{a}}\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\bgroup
\fixFloatSize{images/366f71ce-39bc-47bf-abc0-3b59c3ef130d-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/366f71ce-39bc-47bf-abc0-3b59c3ef130d-upicture1.png}{\includegraphics{images/366f71ce-39bc-47bf-abc0-3b59c3ef130d-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Effect of \textit{Sida acuta} activity on changes in body weight of control and experimental groups}}
\label{f-c756028815cf}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/536da7a9-b416-480b-8593-ff803d5b8b40-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/536da7a9-b416-480b-8593-ff803d5b8b40-upicture2.png}{\includegraphics{images/536da7a9-b416-480b-8593-ff803d5b8b40-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Effect of ethanolic extract of \textit{Sida acuta} on serum urea, uric acid and creatinine in experimental rats }}
\label{f-bee3281bead4}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/8e17a216-5430-4c5a-b626-2a9452c1da80-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/8e17a216-5430-4c5a-b626-2a9452c1da80-upicture3.png}{\includegraphics{images/8e17a216-5430-4c5a-b626-2a9452c1da80-upicture3.png}}{}
\makeatother 
\caption{\boldmath {Effect of ethanolic extract of \textit{Sida acuta} on serum glucose in experimental rats}}
\label{f-4945c9cd07c8}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/2d30c88d-a0de-45d0-adfb-6beb0803f029-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/2d30c88d-a0de-45d0-adfb-6beb0803f029-upicture4.png}{\includegraphics{images/2d30c88d-a0de-45d0-adfb-6beb0803f029-upicture4.png}}{}
\makeatother 
\caption{\boldmath {Effect of ethanolic extract of \textit{Sida acuta } on serum protein in experimental rats}}
\label{f-b91720948d04}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/4e7154e8-f507-40fb-b2c9-ba5b685fe7a6-upicture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/4e7154e8-f507-40fb-b2c9-ba5b685fe7a6-upicture5.png}{\includegraphics{images/4e7154e8-f507-40fb-b2c9-ba5b685fe7a6-upicture5.png}}{}
\makeatother 
\caption{\boldmath {Effect of ethanolic extract of \textit{Sida acuta} on serum AST \& ALT in experimental rats}}
\label{f-4f0a3e7cb343}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/134ee68e-9ee0-4b8b-b9eb-0af60dc85e5c-upicture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/134ee68e-9ee0-4b8b-b9eb-0af60dc85e5c-upicture6.png}{\includegraphics{images/134ee68e-9ee0-4b8b-b9eb-0af60dc85e5c-upicture6.png}}{}
\makeatother 
\caption{\boldmath {Effect of ethanolic extract of s\textit{ida acuta} on serum HDL and cholesterol in experimental rats }}
\label{f-ab812029d539}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/d70f5bfc-a539-4d99-806c-39387741deba-upicture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d70f5bfc-a539-4d99-806c-39387741deba-upicture7.png}{\includegraphics{images/d70f5bfc-a539-4d99-806c-39387741deba-upicture7.png}}{}
\makeatother 
\caption{\boldmath {Histopathology of liver. A {\textendash}Control, B {\textendash} 100mg/kg BW, C {\textendash} 250mg/kg BW, D {\textendash} 500mg/kg BW, E {\textendash}1000mg/kg BW, F{\textendash} 2000mg/kg BW(Magnification 40x)}}
\label{f-560ae1f586f4}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/8eca27aa-6fc1-4d0f-a452-8b5536a36bb4-upicture8.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/8eca27aa-6fc1-4d0f-a452-8b5536a36bb4-upicture8.png}{\includegraphics{images/8eca27aa-6fc1-4d0f-a452-8b5536a36bb4-upicture8.png}}{}
\makeatother 
\caption{\boldmath {Histopathology of kidney. A {\textendash}Control, B {\textendash} 100mg/kg BW, C {\textendash} 250mg/kg BW, D {\textendash} 500mg/kg BW, E {\textendash}1000mg/kg BW, F{\textendash} 2000mg/kg BW(Magnification 40x)}}
\label{f-f0475727477d}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/fc7bcfec-743a-40cb-8b4a-2c1fe869820c-upicture9.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/fc7bcfec-743a-40cb-8b4a-2c1fe869820c-upicture9.png}{\includegraphics{images/fc7bcfec-743a-40cb-8b4a-2c1fe869820c-upicture9.png}}{}
\makeatother 
\caption{\boldmath {Histopathology of colon. A {\textendash}Control, B {\textendash} 100mg/kg BW, C {\textendash} 250mg/kg BW, D {\textendash} 500mg/kg BW, E {\textendash}1000mg/kg BW, F{\textendash} 2000mg/kg BW(Magnification 40x)}}
\label{f-31f48407c8a5}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/66300033-5e57-4479-ace7-73a8087df952-upicture10.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/66300033-5e57-4479-ace7-73a8087df952-upicture10.png}{\includegraphics{images/66300033-5e57-4479-ace7-73a8087df952-upicture10.png}}{}
\makeatother 
\caption{\boldmath {Body weight changes in sub-chronic toxicity study}}
\label{f-62f8ab7111ca}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/77eeed25-a806-4f2e-a02a-48e99e7fd544-upicture11.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/77eeed25-a806-4f2e-a02a-48e99e7fd544-upicture11.png}{\includegraphics{images/77eeed25-a806-4f2e-a02a-48e99e7fd544-upicture11.png}}{}
\makeatother 
\caption{\boldmath {Effect of plant extract on Serum urea, uric acid and creatinine level of control and experimental rats}}
\label{f-df0c8dd47150}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/ced82a82-fded-46a0-8fa9-e04726ad6911-upicture12.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/ced82a82-fded-46a0-8fa9-e04726ad6911-upicture12.png}{\includegraphics{images/ced82a82-fded-46a0-8fa9-e04726ad6911-upicture12.png}}{}
\makeatother 
\caption{\boldmath {Effect of plant extract on Serum Glucose level of control and experimental rats}}
\label{f-5b5e9a625eeb}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/a7731c27-de42-4661-9d6a-863c97327b8b-upicture13.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/a7731c27-de42-4661-9d6a-863c97327b8b-upicture13.png}{\includegraphics{images/a7731c27-de42-4661-9d6a-863c97327b8b-upicture13.png}}{}
\makeatother 
\caption{\boldmath {Effect of plant extract on Serum protein level of control and experimental rats}}
\label{f-df5a641d0d6d}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/98727e82-9a74-4002-b00a-2f28142655f0-upicture14.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/98727e82-9a74-4002-b00a-2f28142655f0-upicture14.png}{\includegraphics{images/98727e82-9a74-4002-b00a-2f28142655f0-upicture14.png}}{}
\makeatother 
\caption{\boldmath {Effect of plant extract on serum cholesterol and HDL of control and experimental rats }}
\label{f-f11ed8b9c044}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/6641ae0a-dc59-4923-ae41-0fd2dfde354e-upicture15.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/6641ae0a-dc59-4923-ae41-0fd2dfde354e-upicture15.png}{\includegraphics{images/6641ae0a-dc59-4923-ae41-0fd2dfde354e-upicture15.png}}{}
\makeatother 
\caption{\boldmath {Effect of plant extract on Serum AST \& ALT level of control and experimental rats}}
\label{f-fc979aeddf24}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/47bd1998-0793-47cf-8b36-efa80714fc04-upicture16.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/47bd1998-0793-47cf-8b36-efa80714fc04-upicture16.png}{\includegraphics{images/47bd1998-0793-47cf-8b36-efa80714fc04-upicture16.png}}{}
\makeatother 
\caption{\boldmath {Histopathology of liver. A {\textendash}Control, B {\textendash} 100mg/kg BW, C {\textendash} 200mg/kg BW, D {\textendash} 300mg/kg BW, E {\textendash}400mg/kg BW, F{\textendash} 500mg/kg BW(Magnification 40x)}}
\label{f-095770d51d93}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/7316efd6-3d6f-4c5d-9a4a-21699c1aa46c-upicture17.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/7316efd6-3d6f-4c5d-9a4a-21699c1aa46c-upicture17.png}{\includegraphics{images/7316efd6-3d6f-4c5d-9a4a-21699c1aa46c-upicture17.png}}{}
\makeatother 
\caption{\boldmath {Histopathology of kidney. A {\textendash}Control, B {\textendash} 100mg/kg BW, C {\textendash} 200mg/kg BW, D {\textendash} 300mg/kg BW, E {\textendash}400mg/kg BW, F{\textendash} 500mg/kg BW(Magnification 40x)}}
\label{f-026b85fe6827}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/0e87b592-d7b5-443d-b03a-4cc8a63642e1-upicture18.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/0e87b592-d7b5-443d-b03a-4cc8a63642e1-upicture18.png}{\includegraphics{images/0e87b592-d7b5-443d-b03a-4cc8a63642e1-upicture18.png}}{}
\makeatother 
\caption{\boldmath {Histopathology of colon. A {\textendash}Control, B {\textendash} 100mg/kg BW, C {\textendash} 200mg/kg BW, D {\textendash} 300mg/kg BW, E {\textendash}400mg/kg BW, F{\textendash} 500mg/kg BW(Magnification 40x)}}
\label{f-c939cb6be2a4}
\end{figure}
\egroup

    
\section{Materials and Methods}
\textbf{Plant collection}

The whole plant \textit{S.acuta} was collected from Tuticorin (DT) and authenticated in Botanical Survey of India, TNAU Campus, Coimbatore (Voucher number: BSI/SRC/5/23/2016/ Tech./348). 

\textbf{Preparation of crude extract}

50g of powdered leaf material was weighed and extracted with 250 ml of ethanol for 72 hrs with occasional shaking. The supernatant was collected and concentrated at 40\ensuremath{^\circ}C. It was stored at 4\ensuremath{^\circ}C in airtight bottles for further studies.\textbf{\space }

\textbf{Experimental Animals}

Healthy Wistar albino rats were procured for the toxicity study. The study was accepted by IAEC (Institutional Animal Ethical Committee) constituted for CPCSEA, Government of India (KAHE/IAEC/Ph.D./163).

\textbf{Experimental procedure}

Acute toxicity assessment was performed for 14 days according to the guidelines as per the Organization for Economic Cooperation and Development (OECD) guidelines 425 \unskip~\citep{798243:19046040}.A total of 18 healthy Wistar albino male rats were divided similarly into six groups of 3 each. Different doses of 100, 250, 500, 1000 and 2000mg/kg BW of ethanolic extract of \textit{S.acuta} were given to Wistar albino rats in the very first day of experimental analysis. Control group receives normal saline and water. They were all placed under observation after treatment for 24h, 48h and 72h for any behavioural, neurological changes and then mortality for the 14 days. Later, the animals were euthanised on the 15th day. Biochemical parameters in serum and histological (liver, kidney and colon) examinations were determined by standard protocols.

For subchronic toxicity study ethanolic leaf extract of \textit{s.acuta} was administrated to a total of 15 albino rats, randomly grouped into five groups of three in each (Group II, III, IV, V and VI) Group I~served as control and was administered with 0.5 ml of normal saline once daily for 28 days. 

Rats in groups II, III, IV, V and VI were orally gavaged with 100, 200, 300, 400 and 500 mg/kg body weight of the ethanolic extract of \textit{Sida acuta } respectively once daily for 28 days. The rats were observed daily for any signs of toxicity, and their body weights were also recorded weekly throughout the experimental period. Then the rats were euthanised on 29\ensuremath{^{th}} day, by keeping overnight fasting for 12 hours prior to euthanasia. Blood samples were collected by cardiac puncture into EDTA and non EDTA coated tubes for haematological and biochemical investigations, respectively. Liver, colon and kidneys were fixed in 10\% formalin for histopathological examination.

\textbf{Biochemical estimation }

Blood samples were collected in non-anticoagulant tubes and centrifuged at 3000 rpm. The serum was separated and used for quantification. Commercially available kits from (Agappe Diagnostics) were used for the biochemical assays such as glucose, protein, urea, uric acid creatinine, cholesterol, HDL, AST and ALT. The haematology autoanalyser determined hematologic parameters like red and white blood cells, haemoglobin, MCV, MCH, MCHC, platelets, lymphocytes, monocytes, eosinophils, neutrophils and basophils.

\textbf{Histopathology}

\textbf{\space }After euthanasia, liver, kidney and colons were poised for the histology study. All the organs were secured in 10\% formalin, fixed in paraffin, divided approximately four \ensuremath{\mu }m width and marked with H\&E (hematoxylin and eosin) for examination under light microscopy.\textbf{\space }

\textbf{Statistical analysis}

\textbf{\space }Results expressed as Mean \ensuremath{\pm} SD were evaluated by one way ANOVA, followed by DMRT comparison(5\% level of significance) using SPSS. 
    
\section{Results and Discussion}
\textbf{Acute Toxicity Study}

\textbf{Behavioural and body weight changes}

The effect of ethanolic extract of \textit{Sida acuta}  Burm.f leaves on the bodyweight of rats is shown in Figure~\ref{f-c756028815cf}. The acute toxicity analysis did not show any signs and symptoms up to 2000 mg/kg BW. No morbidity or mortality was observed in the treated groups in all doses during the experimental period. The body weights were measured on 1\ensuremath{^{st}}, 7\ensuremath{^{th}} and 14\ensuremath{^{th}} day. As per observation, an initial gradual decrease in body weight in all groups were observed on 7\ensuremath{^{th}} day and increased to normal on 14\ensuremath{^{th}} day. So it can be assumed that the LD\ensuremath{_{50}} of the extract could be greater than 2000 mg/kg body weight. 

\textbf{Effect of ethanolic extract of \textit{Sida acuta} on serum biochemical parameters}

In the present study, biochemical parameters urea, uric acid and creatinine in the treated groups 250mg/kg BW(Urea), 100 \& 500mg/kg BW(Uric acid) and 250 \& 500mg/kg BW(creatinine) shows statistically significant variation from control groups and the remaining groups are similar to that of control(Figure~\ref{f-bee3281bead4}). Even though there is significant difference in groups (250mg/kg BW(Urea), 100 \& 500mg/kg BW(Uric acid) and 250 \& 500mg/kg BW(creatinine), the values are pertaining to the normal range of experimental animals. So it can be considered as safe as for as kidney markers as considered.

Figure~\ref{f-4945c9cd07c8} shows that there is a statistically significant change in blood glucose level in 500mg/kg BW treated rats, and the remaining treated groups are similar to that of the healthy control rats. As the statistically significant difference comes under the normal range and hence cannot be treated as toxicity. 

In the experimental study, the level of protein in treated groups (500 mg/kg BW \& 2000 mg/kg BW) shows a slight deviation from control rats (Figure~\ref{f-b91720948d04}). The slight elevation of protein level pertains within the normal range. So it can be suggested that there are no signs of toxicity.

Figure~\ref{f-4f0a3e7cb343} shows that there are statistically significant changes in AST and ALT levels in treated groups compared with the control rats. Though there is a variation, the alterations are within normal range and hence cannot be counted as an implication of toxicity. In the present study, the level of cholesterol and HDL in the treated doses 250 \& 500 mg/kg BW (cholesterol) and 1000 mg/kg BW (HDL) shows statistically significant variation when compared to the control rats and other remaining treated groups are showing typical values as that of control rats (Figure~\ref{f-ab812029d539} ). The slight elevation of HDL and cholesterol within the normal range. So it may not be considered as a sign of toxicity.

\textbf{Histopathological examination}

Microscopic examination of liver, kidney and colon displayed in Figures~\ref{f-560ae1f586f4}, \ref{f-f0475727477d} and~\ref{f-31f48407c8a5} showed no histopathological changes in both control and the treated groups. Thus the histopathological examination supports the protective nature of \textit{Sida acuta } on the liver. Study of morphological and anatomical features of kidney tissues of the rats showed the standard architecture of the kidney with glomeruli and tubules signifying that the drug is devoid of nephrotoxic effect. All the control and treated animals showed the standard architecture of the colon. 

\textbf{Sub chronic toxicity study}

\textbf{Bodyweight changes}

Neither death, no toxicity signs were observed on animals following 28 days of treatment with various doses of \textit{Sida acuta } (100 -500mg/kg BW). From the 0\ensuremath{^{th, }} 14\ensuremath{^{th}} and 28\ensuremath{^{th}} day of treatment, a gradual increase in body weight gain was noted in all experimental animals, except control group (Figure~\ref{f-62f8ab7111ca} ).

\textbf{Haematological Parameters}

The effect of the ethanolic extract of \textit{Sida acuta} on haematological indices was examined at the end of treatment (Table~\ref{tw-030d44184b0b}). Results showed no significant variance on haematological parameters of experimental groups compared to control. Indicates that the plant extract is devoid of toxic substance that can cause anaemia or other defects.

\textbf{Effect of ethanolic extract of \textit{Sida acuta} on serum biochemical parameters}

Results of the biochemical parameters (urea, uric acid and creatinine) are shown in Figure~\ref{f-df0c8dd47150}. The level of uric acid in the following groups 100, 300 and 500 mg/kg BW exhibited a significant increase from control and 200 \& 400 mg/kg BW found to be statistically similar to that of control indicating the efficacy of respective doses. The statistical variation mentioned above falls in the normal range and hence no evidence of toxicity.

 In the present sub chronic toxicity study, statistical change in blood glucose level in treated doses 300 \& 500 mg/kg BW from control rats and doses 200 \&400mg/kg BW shows no statistically significant variation with control rats (Figure~\ref{f-5b5e9a625eeb} ). The slight elevation level of experimental groups is pertaining within the normal range. So we can consider it as no evidence of toxicity.

 In the present study, the level of protein is slightly increased in 300 mg/kg BW treated rats, and all the other groups show normal range as that of control (Figure~\ref{f-df5a641d0d6d}). The slight elevation is also within the normal range we can consider it as devoid of toxicity as far as protein is considered.

In the level of cholesterol and HDL in the treated doses 200 \& 400mg/kgbw (cholesterol) and 300 \& 500mg/kgbw(HDL) shows statistically significant variation when compared to the control rats and other remaining plant treated groups are no statistically significant variation from control rats (Figure~\ref{f-f11ed8b9c044}). The slight elevation of HDL and cholesterol are within the normal range. So it cannot be considered as a sign of toxicity. 

Figure~\ref{f-fc979aeddf24} shows that there is a statistically significant change in AST (300 \& 500mg/kg BW) and ALT (200 mg/kg BW) levels in treated groups compared with the control rats and the remaining plant extract groups show statistically no significant variation from control rats. But the alterations are within the control range and hence cannot be counted as an implication of toxicity.

\textbf{Histopathology of liver, kidney and colon}

Histopathological examination of the liver, kidney and colon sections showed normal histology of these organs at all treatment doses compared to the control group. No toxic effects on the histological appearances of the liver, kidney and colon of the treated rats (Figures~\ref{f-095770d51d93}, \ref{f-026b85fe6827} and~\ref{f-c939cb6be2a4}) were noticed.

Toxicity is an expression of being poisonous, indicating the state of adverse effects led by the interaction between toxicants and cells \unskip~\citep{798243:19046036,798243:19046233}. Herbal constituent based drugs or herbal formulations, in general, have low toxicity index compared to chemical drugs. Toxicological evaluation is a preliminary step in pharmacology to evaluate whether the respective plant has any altered modulations of biochemical and metabolic interactions. For acute toxicity study, animals did not exhibit any signs of toxicity for doses up to 2000 mg/kg BW in single-dose administration. No mortality and no changes in behavioural parameters were observed. Bodyweight parameters also fall in the normal range without any expression of toxicity.

Acute toxicity aimed to assess whether the respective extract is toxic in dangerous doses. From the study, it can be concluded that \textit{S.acuta} ethanolic extract can be safely administrated orally up to 2000mg/kg BW. Sub chronic evaluation is significant because, in most of the diseases, there is a requirement for repeated administration of medications. Hence it is relevant to validate whether the plant sample has any toxicity in repeated dosage formulations in \textit{in vivo} models. Sub chronic toxicity study examines toxicity caused by repeated dosing over an extended period of 28 days of oral administration in rats. This test illuminates whether the repeated dosage implementation provides any changes in biochemical and haematological parameters and any histological alterations in target vital organs. The results are convincing in having no observable toxicity in any of the parameters analysed.

In toxicological evaluation, biochemical parameters have significant roles as a marker because of their immediate response to toxicants and associated clinical signs and symptoms. The results on biochemical parameters have shown a statistically significant difference in relevant parameters such as glucose, AST, ALT, Urea, Uric acid, creatinine, cholesterol, HDL, and total protein in experimental groups compared to control rats. But observed variations of various parameters come under the normal range so it can be inferred that plant extract has got no toxic effect on serum biochemical parameters. Serum transaminases are relevant markers of hepatic toxicity because of their immediate response to hepatic infection or injury. In the present analysis, enzymatic markers don't show any alterations in the toxic range \unskip~\citep{798243:19046044}.

An alteration in haematological parameters in treated rats is an indication of anaemia and erythropoiesis. In the present study, the haematological parameters are about the normal range is an indication of the safety nature of the plant extract. Bodyweight change is an essential index for assessment of toxicity as normal growth directly correlates with normal metabolism \unskip~\citep{798243:19046045}. 

In the present observation, body weight changes don't indicate any sign of toxicity. In the present study, all biochemical parameters remain relatively constant changes. Measurement of blood serum urea has been used for many years as an indicator of kidney marker \unskip~\citep{798243:19045979,798243:19046037}. 

In the present study, the mean values of urea, uric acid and creatinine are in the same range as that of control groups. AST and ALT are found primarily in the liver and is the most sensitive marker for liver cell damage. In this study, the mean values of AST and ALT also fall in the normal range as that of control. Histological examination of treated and control rats showed that ethanolic extract \textit{of S.acuta} does not cause any toxic and noted histopathological changes on kidney, liver and colon. Based on the results of the present study, it can be concluded that leaves of \textit{S.acuta} ethanolic extract is being non-toxic for internal administration in animal models and could be well used for pharmacological and therapeutic purposes. Based on the sub chronic toxicity evaluation the lowest effective dose of 200mg/kg BW and a comparatively higher dose of 400mg/kg BW was determined for further in\textit{ vivo} anticancer study of \textit{Sida acuta} burm,f ethanolic leaf extract in Wistar albino rats.
    
\section{Conclusion}
The acute toxicity study of the ethanolic extract of the \textit{s.acuta} leaves did not produce adverse effects on behaviour changes and gross pathology of the rats up to 2000mg/kg BW. Hence lethal dose for \textit{s.acuta} is above 2000mg/kg BW. Sub chronic toxicity study of the ethanolic extract of the \textit{s.acuta} leaves did not have any adverse effect on body weight, biochemical and hematological parameters at tested doses. Kidney, liver and colon tissues exhibit no sign of toxicity. From sub chronic toxicity study, the doses 200 and 400mg/kg BW as lowest and highest effective dose respectively was standardised for further research in animal models using the individual plant extract. 
\section*{Acknowledgement}The authors are thankful to our Chancellor, Chief Executive Officer, Vice-Chancellor and Registrar of Karpagam Academy of Higher Education for providing facilities and encouragement to complete this research work in exceptional fulfilment. 

\textbf{Funding Report}

None

\textbf{Conflict Of Interest}

The authors declare no conflict of interest 


    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
